Author’s response to reviews

Title: Novel Deletion Alleles Carrying CYP21A1P/A2 Chimeric Genes In Brazilian Patients With 21-Hydroxylase Deficiency

Authors:

Fernanda B Coeli (fbcoeli@bol.com.br)
Fernanda C Soardi (soardi@unicamp.br)
Renan D Bernardi (renandarin@gmail.com)
Marcela de Araújo (maraujo@boldrini.org.br)
Luciana C Paulino (luciana.paulino@ufabc.edu.br)
Ivy F Lau (mpmello276@hotmail.com)
Reginaldo J Petroli (rpetroli@unicamp.br)
Sofia HV Lemos-Marini (sofia.lemos@terra.com.br)
Maria TM Baptista (terma@uol.com.br)
Gil Guerra-Júnior (gileandrea@uol.com.br)
Maricilda Palandi de Mello (mmello@unicamp.br)

Version: 5 Date: 7 June 2010

Author’s response to reviews: see over
June 7th, 2010.

To
Scott Edmunds PhD
The BioMed Central Editorial Team
e-mail: editorial@biomedcentral.com
Tel: +44 (0) 20 3192 2013

MS: 1002574954320354 Formatting changes

Dear Dr. Edmunds,

This is to submit the formatted manuscript entitled “Novel Deletion Alleles Carrying CYP21A1P/A2 Chimeric Genes In Brazilian Patients With 21-Hydroxylase Deficiency” to be published in BMC Medical Genetics.

The manuscript has been revised. We have addressed major requests as follow:

Major revisions

Figures: Please reduce the volume of text presented each figure legend to no more than 250 words or it will compete with the image for space in the final publication.

We have changed figure 3 legend to 251 words.

We thank you for your attention.

Yours sincerely,

Maricilda Palandi de Mello, Ph.D.